Author's response to reviews

Title: Domestic violence and mental health in Bolivia

Authors:

  Dominique Meekers (dmeekers@tulane.edu)
  Sarah C Pallin (sarah.c.pallin@gmail.com)
  Paul Hutchinson (phutchin@tulane.edu)

Version: 2  Date: 25 September 2012

Author's response to reviews:

In response to the editor's feedback, we made the following changes/clarifications:

1. Ethics: We clarified that the data used in this paper were collected by Macro International as part of the larger Demographic and Health Surveys project, and we clarified the ethics approval process (which we obtained from Macro). These changes are on page 3 of the revised manuscript.

2. Questionnaire: We included a hyperlink to the survey questionnaire source. (Also on page 3 of the revised manuscript).